THE QUALITY OF ENVIRONMENTAL DISCLOSURE IN VARIOUS REPORTING MEDIA OF OIL AND GAS COMPANIES IN DEVELOPING COUNTRIES

Abdullah Hamoud Ismail*, Azhar Abdul Rahman**

* Ministry of Oil and Minerals, Sana`a, Yemen ** School of Accountancy, College of Business, University Utara Malaysia, Kedah, Malaysia

Abstract

Corporate environmental reporting (CER) plays important role due to the increase in public awareness of environmental issues. Hence, to be beneficial, corporate managers should not merely display CER information but rather emphasize on the quality of information disclosed. The quality of CER can be seen as a key value for companies and many benefits could be provided if companies released high quality environmental information. Prior environmental disclosure literature has not focused much on disclosure quality; instead, it concentrated on the quantity of disclosure. In addition, most of the few studies that focused on quality of environmental disclosure have revealed low level of quality of such disclosure. Therefore, this study aims to investigate the quality of environmental disclosure in different reporting mediums by oil and gas companies in developing countries. Using content analysis, an index and scoring scheme were applied to the annual reports, stand-alone reports and corporate homepages of a sample of 116 oil and gas companies in 19 developing countries. The results of this study reveal that the quality of the environmental disclosure of the sample companies is relatively high compared to previous studies. This study has important implications in enhancing the understanding of environmental disclosure practices of oil and gas companies in developing countries.

Keywords: Environmental Disclosure Quality, Reporting Media, Oil and Gas Companies, Developing Countries

1. INTRODUCTION

Environmental issues have increasingly drawn the attention of the world at different levels, and corporate social and environmental responsibility has become a major contemporary focus of business, government and community attention globally (Parker, 2014). As a result, interest in corporate disclosure of environmental information has grown in recent years (Rupley, Brown and Marshall. 2012). However, attention on environmental disclosure (ED) has been confined to the companies of developed countries, while the developing countries suffer from environmental disclosure practices in corporations (Eljayash et al., 2012; Kaur, 2015).

Among the largest consumers of natural and social resources, business organizations have come under increased pressure to justify the nature and scale of their consumption. Specifically, business organizations, particularly industrial communities, are considered more and more as responsible for their impacts to the environment and society (Brammer & Pavelin, 2006). The oil and gas industry is among the industries with the greatest impacts on the environment. According to the International

Energy Agency (IEA, 2015), energy-related carbon dioxide (CO₂) emissions are the majority of global greenhouse gas (GHG) emissions, while, oil and gas are the largest source of fuel combustion emissions and responsible for approximately 53% of global energy-related CO₂ emissions in 2013. It is well recognized that environmental effects of the petroleum operations on the natural environment are very high (Mughal, 2014). In addition to the environmental effects that result from normal operations of oil and gas activities, the effects may be the results of occasional events such as, oil spill and explosion. During the last four decades, the oil and gas industry has witnessed several critical environmental incidents, for example; Exxon Valdez oil spill of Alaska in 1989, and Gulf of Mexico oil spill of 2010. The occurrences of environmental incidents as a result of activities of oil companies have contributed to the increase of environmental awareness and put the oil and gas industry under societal pressure to reduce its impacts on the environment (Eljayash et al., 2012; Frynas, 2009; Islam and Islam, 2011). However, the increase of environmental awareness has largely influenced businesses to engage in environmental management and practice including environmental reporting (Yusoff and Othman, 2013).

A large volume of the world's proven recoverable reserves of crude oil and natural gas liquids is held by the DCs, and most of production is also produced by them. At the end of 2013, the DCs held 82% of the world`s proven recoverable reserves of oil and natural gas liquids and accounted for 67% of world's production of oil and natural gas liquids (Eni`s World Oil and Gas Review, 2014). Considering this significant amount of reserves and production of oil and gas coupled with the environmentally sensitive nature of this industry, makes the DCs highly exposed to environmental impacts. addition, it was recognized that, in the era of the worldwide globalization, presence multinational companies and highly publicized environmental incidents in developing countries, issues of corporate social responsibility (CSR) and its publications seem to be more significant in developing nations (United Nations Research Institute for Social Development [UNRISD], 2000). However, it was recognized that the adverse effects of the oil and gas companies are greater in the developing countries (Abdalla and Siti-Nabiha, 2015). It is also recognized that the success of operations of multinational companies (MNCs) in host countries can be greatly impacted by their level of local acceptability, and occurrence of major oil disasters raise a question as to how international companies can effectively manage local expectations and the associated problems of oil production in order to gain local acceptability (Fragouli and Danyi, 2015). Companies use environmental disclosure as a mechanism to manage society expectations toward corporate operations and increase reputations (Haji, 2013; Perez, 2015; Yin, 2012). All these make reporting on environmental aspects of oil and gas companies very important from the governments and preparers' (companies) point of views.

However, the literature points out that the majority of previous studies concerned with social and environmental reporting have been conducted in the developed world, but comparatively limited studies have been undertaken in the developing countries (Eljayash, Kavanagh and Kong, 2013; Joseph, Pilcher and Taplin, 2014; Kansal, Joshi and Batra, 2014; Kaur, 2015; Lu and Abeysekera, 2014; Mughal, 2014; Yusoff and Othman, 2013). In practice, social and environmental disclosure has matured in some developed countries; however, in some developing nations, it is still a relatively new practice (Kaur, 2015; Lu and Abeysekera, 2014; Mughal, 2014). In particular context of a developing country, Djajadikerta and Trireksani (2012) indicated that the practice of corporate social and environmental disclosure (CSED) in Indonesia is still at an early stage, and most of the companies still have a lack of understanding about CSED. Ahmad and Hossain (2015) concluded that disclosure of climate change and global warming in the annual reports of Malaysian companies is still at its introductory stage. In addition, it was noted that findings of studies that focused on the developed countries cannot be generalized to less developed countries as differences in culture and nationality are expected to influence the accounting and environmental practices (Matthew, 1993; Perera & Matthews, 1990).

It is noted that even though awareness of the

sustainability and social responsibility issues has grown, sustainability and social responsibility reporting has long been voluntary and it has not been regulated by legislations, as a result, many of the companies still do not disclose on such issues and Sticks, 2013; Vuorela, Specifically, environmental disclosure worldwide is generally unregulated and voluntary in nature, (De Villiers and Van Staden, 2012; Michelon, Pilonato and Ricceri, 2015; Sen, Mukherjee and Pattanayak, 2011). As environmental disclosure is primarily voluntary, companies are free to choose what and how to disclose (Ahmed & Sulaiman, 2004; Peiyuan, 2005). This causes quality problems such as consistency and comparability, both over time and between companies (De Villiers and Van Staden, 2012).

Even though in some countries companies are mandated to disclose information on their CSR aspects, the mandatory requirements of such disclosures do not detail specific information to be disclosed by companies. Instead, companies are given the flexibility to provide information relating to their CSR activities (Haji, 2013). Thus, the lack of specific, formal national and international regulations seems to allow companies much flexibility in how they carry out their social and environmental reporting activities and allow them to use guidelines in a biased manner (Haji, 2013; Michelon et al., 2015). As a result, there is a lack of completeness in environmental disclosure (Michelon et al., 2015), and environmental disclosure varies substantially in terms of content, information, graphic information and length (Said, Omar and Abdullah, 2013).

It was argued that measuring the quality of disclosure is important, and that investigating only the volume of disclosure can be misleading (Hassan, 2010; Hooks and van Staden, 2011), as evaluating the quality of disclosures adds a further dimension to the assessment of environmental reporting (Hooks and van Staden, 2011). Despite this, majority of previous studies concerned with environmental disclosure concentrated only on the quantity of disclosure but scant attention has been given to the quality of such disclosure (Aburaya, 2012; Ahmad and Haraf, 2013; Chatterjee and Mir, 2008; Cuesta and Valor, 2013; Eltaib, 2012; Hassan, 2010; Haji, 2013; Michelon et al., 2015; Rupley et al., 2012; Sulaiman et al., 2014).

From literature review, it is noted that, with the exception of a few studies (e.g. Aburaya, 2012; Ahmad and Haraf, 2013; Ane, 2012; Belal, 2000; Brammer & Pavelin, 2006, 2008; Comyns and Figge, 2015; Cormier, Magnan & Van Velthoven, 2005; Cuesta and Valor, 2013; Darus, Hamzah and Yusoffa, 2013; Dong, Fu, Gao and Ni, 2015; Eakpisankit, 2012; Eljayash, 2015; Eljayash et al., 2012; Haji, 2013; Hassan, 2010; Harun, Abdul Rashid and Alrazi, 2013; Hooks & Van Staden, 2011; Lu et al., 2015; Michelon et al., 2015; Oba and Fodio, 2012a; Rupley et al., 2012; Sulaiman et al., 2014; Wiseman, 1982), who focus on disclosure quality, the social and environmental disclosure in previous studies were not able to capture the quality of the disclosure.

In addition, prior studies on social and environmental disclosure quality suffer from methodological limitations as most of these studies used disclosure quantity measures to assess the quality of disclosure. It was argued that the disclosure instruments used in previous social and environmental disclosure studies have been built primarily on a checklist of items that capture the amount and variety of disclosure. This approach was criticized as it does not sufficiently determine the quality of information (Michelon et al., 2015). According to Eakpisankit (2012) in spite of some previous studies that have claimed to measure the quality of disclosures, in reality they have merely focused on content rather than identifying concrete qualitative concepts.

Aburaya (2012) stated that, the assessment of environmental disclosures quality remains a rather controversial issue. Sulaiman et al. (2014) stressed that the quality of environmental information reported should be considered. However, until now, researchers still opined that there is a scarce of literature regarding social and environmental disclosure quality (Michelon et al., 2015). Thus this study contributes to fill this gap in literature by devising a conceptual framework of environmental disclosure quality that is dependent upon the quality of environmental disclosure rather than its amount or extent.

In terms of industry, the oil and gas industry is among the industries with the greatest impacts on environment (IEA, 2015). The environmental effects on the petroleum operations on the natural environment are very high, as the operations of this industry cause air pollutions and responsible for the waste they emit in the sea which is very disastrous for the life under sea (Mughal, 2014). The oil and gas industry is considered a main source of environmental problems, as its operations involve many potential negative environmental effects (Ariweriokuma, 2009; Frynas, 2009). The occurrences of environmental incidents as a result of activities of companies, including oil companies, have contributed to the increase of environmental awareness in many countries over the world (Eljayash et al., 2012; Hossain et al., 2006; Frynas, 2009; Sustainability & UNPE, 1999).

There are a few studies that examined environmental disclosure in oil and gas industry (cf. Alciatore and Dee, 2006; Al-Drugi and Abdo, 2012; Barr, 2007; Bose, 2006; Dibia and Onwuchekwa, 2015; Eljayash et al., 2012; Eljayash et al., 2013; Guenther, Hoppe and Poser, 2007; Heflin and Wallace, 2014; Oba and Fodio, 2012b; Patten, 1992; Summerhays and De Villiers, 2012; Sustainability Ltd. & UNEP, 1999). It was argued that, in order to enhance our understanding on environmental disclosure behavior, it is important to focus on a specific industry (Gray et al., 1995; Ahmad and Haraf, 2013). However, the adverse effects of oil and gas companies are greater in the developing countries (Abdalla and Siti-Nabiha, 2015). All these motivate to conduct a study focusing environmental disclosure of oil and gas industry in developing world.

Moreover, with the exception of Oba and Fodio (2012a) and Eljayash et al. (2012), no studies have analyzed the quality of environmental disclosure in oil and gas industry. However, these two studies are suffering from some limitations, such as limiting themselves to annual reports and the samples are small. It was argued that, in order to enhance our understanding on environmental disclosure

behavior, it is important to focus on a specific industry (Gray et al., 1995; Ahmad and Haraf, 2013). Hence, this study intends to fill this knowledge gap by examining the quality of environmental disclosure made by oil and gas companies in developing countries

The current study attempts to fill the gaps in the literature by examining corporate environmental disclosure quality (rather than its quantity), concentrating on environmental disclosure made in the three main mediums of reporting (namely, annual reports, stand-alone reports and corporate homepages). Precisely, this study aims to determine the level of environmental disclosure quality of oil and gas companies in developing countries.

Prior environmental disclosure literature has little focused on disclosure quality; instead, it concentrated on the quantity of disclosure. The current study seeks to fill the gap in the literature by considering the issue of environmental disclosure quality (rather than quantity). Assessing the quality of the environmental reporting enables an identification of the strengths and weaknesses in current reporting practice and advances our understanding. In addition, contrary to the most available literature that only focuses on sole medium of environmental disclosure (mostly annual report), the current study contributes to the literature by covering most common vehicles of environmental disclosure, particularly, annual reports, stand-alone reports and corporate homepages. This study also fills the void in prior environmental disclosure literature regarding whether various reporting mediums vary regarding their disclosure quality.

The present study also contributes to the environmental disclosure literature by centering on the ED practices of specific sector (i.e. the oil and gas industry) in the DCs. It was argued that, in order to enhance our understanding on environmental disclosure behavior, it is important to focus on a specific industry (Gray et al., 1995; Ahmad and Haraf, 2013). Thus, this study contributes to environmental disclosure literature as it provides insight into the environmental disclosure practices of oil and gas companies within developing countries, where there are limited published studies.

The rest of the paper is structured as follows; section 2 provides the literature review. Section 3 explains the research methodology. Section 4 presents the results from empirical analyses and discussion. Section 5 presents the concluding remarks, implications of the study while section 6 explains the limitations of the study and provides suggestions for further research.

2. LITERATURE REVIEW

Consistent with the increase in public concern of environmental issues, environmental accounting (EA) practice has become an attractive area of research and received attention from many researchers worldwide (Eltaib, 2012). Disclosing information relating to environmental aspects is considered as one of significant issues in relation to environmental accounting (Eltaib, 2012). The current study attempts to investigate the issues of quality of environmental disclosure in various reporting mediums of environmental information.

According to Islam et al. (2005) environmental

disclosure is "an umbrella term that describes the various means by which companies disclose information on their environmental activities". Lodhia (2006)has defined the Corporate Environmental Reporting (CER) as "a process which "companies often disclose environmental information to their stakeholders to provide evidence that they are accountable for their activities and the resultant impact on the environment". Environmental disclosure is also defined by Kuo and Chen (2013) as "a set of information items that relate to a firm's past, current, and future environmental management activities and performance" (p. 1467), and by Yusoff and Othman (2013) as "any written passage about company's environmental issue and activity" (p. 1720). For the purpose of this study, environmental disclosure is defined as a process of communicating the information on environmental issues through various reporting mediums including; annual report, separate stand-alone environmental-related reports (i.e. environmental report, social responsibility report, sustainability report), and corporate homepage of Internet.

Firms use social and environmental disclosures to improve their image in the eyes of different stakeholder groups and public in general and in turn gain their legitimacy for existence (Hossain, Al Bir, Tarique and Momen, 2016; Khlif, Guidara and Souissi, 2015; Kuo and Chen, 2013; Noodezh and Moghimi, 2015; Suchman, 1995). Several earlier studies revealed that firms, especially operating in environmentally sensitive industries, disclose social and environmental information to promote/ enhance their images and reputations and in turn for the legitimization of their societal existence (e.g. Deegan and Gordon, 1996; Deegan and Rankin, 1996; Deegan, Rankin, & Tobin 2002; Khlif et al., 2015; Kuo and Chen, 2013; Neu et al., 1998; Patten, 1992; Yusoff and Lehman, 2009). Thus, social and environmental disclosure is considered a tool which could help companies to influence society's perceptions toward corporate operations (Haji, 2013).

As mentioned previously, the public concern of environmental issues has increased, and as a result environmental accounting practice has received attention from the scholars in the area of accounting research, and much of this research was dominated by studies focused on environmental disclosure (Eltaib, 2012). The majority of prior environmental disclosure studies have focused on the quantity of disclosure but scant attention has given to disclosure quality (Aburaya, 2012; Ahmad and Haraf, 2013; Cuesta and Valor, 2013).

Disclosure quality measure enables to evaluate meaning and importance of disclosure, rather than just the volume (Walden and Schwartz, 1997). Reporting quality can have a significant influence on the quality of the decisions made by stakeholders (Brink et al., 1997). Hasseldine et al. (2005) suggested that environmental disclosure quality as opposed to just quantity has a significant impact on the development of environmental reputation among stakeholder groups of investors and executives.

It was argued that quality reporting does not entail only volume but it should also allow stakeholders to carry out informed decisions that are significant to their intentions (Brink et al., 1997). A primary issue in the context of reporters is the report content; in other words, what makes a really significant issue in the user's viewpoint (Barr, 2007). So, reporting quality should be considered because the failure to encapsulate the content of the environmental information constitutes a failure to cover the issue, its importance and the communicated meanings (Silva, 2008). The quality of the environmental disclosure can be seen as a key value for companies, and many benefits could be provided if the company released high quality environmental information (Rattanaphaphtham and Kunsrison, 2011). It is recognized that the quality of environmental reporting (as compared to its quantity) is important (Sulaiman et al., 2014).

However, prior research revealed that companies disclose a limited amount and poor quality of social and environmental information. For example, Harte and Owen (1991) examined the level of environmental disclosure made by 30 UK companies in their annual reports. The study indicated that, although, the level of environmental disclosure increased during the period, it was not in detail. Choi (1999) examined environmental disclosure in semi-annual reports for 64 Korean companies. The results revealed that the level of environmental disclosure is low.

During the 2000s decade, many studies relating social and environmental disclosure were conducted, and most of them have indicated a low level of quantity and/or quality of social and environmental disclosure. For example, Imam (2000) examined social disclosure of Bangladeshi listed companies. The study revealed that the social disclosure level was very poor and inadequate. Belal (2000) investigated the environmental reporting in Bangladesh. The study revealed that the quantity and the quality of environmental reporting is an inadequate and poor. Belal (2001) investigated the social responsibility disclosure in Bangladesh. He concluded that the level of social and disclosure of Bangladeshi companies is very low and descriptive in nature. Similarly, Belal (2008) revealed that Bangladeshi companies are disclosing social and environmental information only on a limited scale. Elijido-Ten (2004) concluded that the majority of environmental disclosures are still confined to the provision of general or vague descriptions.

Kamla (2007) examined the volume, quality and nature of social reporting practices in the annual reports of 68 companies from nine Arab Middle East countries, namely, Bahrain, Egypt, Jordan, Kuwait, Oman, Qatar, Saudi Arabia, Syria, and United Arab Emirates (UAE). The study revealed that only 10 companies, 15% of the sample, provided some form of environmental information. In addition, most disclosed information related to employee issues, while, the level of disclosure in relation to the environmental dimension the lowest.

Rizk et al. (2008) examined the extent of social and environmental reporting made by Egyptian manufacturing companies in their annual reports. The study indicated that the extent of CSR reporting is low and descriptive in nature. Silva (2008) examined voluntary environmental reporting in the annual reports of New Zealand and Australian publicly listed companies. The study revealed that the level of voluntary environmental reporting in the annual reports of New Zealand and Australian

publicly listed companies is low and demonstrates poor content-quality. Said et al. (2009) examined extent of corporate social responsibility disclosure of Malaysian public listed companies. The results indicated that the level of corporate social responsibility disclosure in Malaysian companies is generally low.

A study of Abd Rahman et al. (2011) was conducted to assess the level of corporate social responsibility disclosure of a sample of governmentlinked companies listed on Bursa Malaysia for the period 2005-2006. They found that the amount of CSR disclosure by Malaysian government link companies to be limited but growing. Liua, Liu, McConkey and Li (2011) investigated environmental disclosure in annual reports and stand-alone environmental and social responsibility reports of steel companies listed in Shanghai Stock Exchange. The study shows significant differences in the form of environmental disclosure, as well as great differences in terms of content and intensity. Djajadikerta and Trireksani (2012) measured the extent of CSED made by Indonesian listed companies on their corporate web sites. They found that the extent of CSED is low and the nature of disclosure is mostly descriptive.

Cuesta and Valor (2013) investigated the quality of environmental, social and governance reporting of Spanish listed companies. They indicated that the sampled companies failed to provide complete information on environmental performance (37%). Harun et al. (2013) examined the quality of sustainability disclosure by 15 commercial banks in Malaysia, and they concluded that the disclosure quality is considered low. Similarly, Darus et al. (2013) revealed that the quality of CSR information disclosed by Malaysian companies on their websites proved to be generally low.

Employing a case study method and using qualitative data, Momin and Parker (2013) investigated social and environmental information disclosed in the annual reports of MNC subsidiaries in Bangladesh. The study concluded that CSRR practice in Bangladeshi MNC subsidiaries is limited. Said et al. (2013) examined the level of environmental disclosure of Malaysian companies. The study revealed that the level of environmental disclosure in Malaysian public listed companies is low.

Ahmad and Haraf (2013) examined the extent, quality, nature and trends of environmental disclosures of Malaysian property development companies. They concluded that companies do not appear to respond to the increased public concern due to recent landslide incidents by increasing the extent or quality of environmental disclosures in their annual reports. Both extent and quality of environmental disclosures are very low and most companies provide mostly soft disclosures. The findings also revealed that companies are not consistent in the extent, nature or quality of environmental disclosures made over time.

Yusoff and Othman (2013) investigated the state of environmental reporting by Malaysian and Australian companies on different mediums. The study revealed that environmental reporting in stand-alone reports (environmental reports, social and sustainability reports), corporate websites, and corporate newsletters is predominantly general and

qualitative in nature. Bowrin (2013) examined the extent to which publicly-listed Caribbean companies provide social and environmental disclosures and the factors related to their disclosure practices. The study revealed that the level of social and environmental disclosure in the Caribbean was relatively low.

Chang (2013) examined the environmental disclosure of listed eclectic companies in China made in their social responsibility reports. The findings indicated that the extent of environmental disclosure is low (with means of 0.1744, 0.1918, 0.1942 and 0.2171 for the years 2008, 2009, 2010 and 2011 respectively). Kamla and Rammal (2013) examined social reporting with special emphasis on themes related to social justice on annual reports and web sites of Islamic banks from 11 countries. The results revealed that social disclosure of the Islamic banks emphasize their religious character through claims that they adhere to Sharia's teachings, but the disclosure lacks specific or detailed information relating to schemes initiatives.

He and Loftus (2014) evaluated the environmental disclosure practices of listed Chinese operating in environmentally sensitive industries, and revealed that, the level of disclosure is low and lag behind that of companies in developed countries. Chithambo and Tauringana (2014) examined the extent of greenhouse gas (GHG) disclosures made in the annual reports, sustainability reports and web sites of London Stock Exchange financial listed companies. The study indicated that the extent of voluntary GHG disclosure of the sample companies is still low.

Joseph et al. (2014) examined extent and determinants of the sustainability reporting in Malaysian local councils' websites. The study indicated that the level of sustainability disclosure on the corporate websites of Malaysian public sector was below average level (26.8%). Kansal et al. (2014) examined level of CSR disclosures made by the top 100 companies in the Bombay Stock Exchange, and found that overall disclosures are low.

Yusoff and Darus (2014) investigated the environmental disclosure practice from an Islamic perspective using content analysis on annual and sustainability reports of Islamic Financial Institutions (IFIs) in Malaysia. The study revealed that the extent of environmental disclosure is low, descriptive and qualitative in nature. The results indicated that the key environmental disclosures provided were related to climate change mitigation and adaptation, and prevention of pollution type of activities. Further exploration on the prioritization of environmental activities found that the key focus of the vital activities was prevention related programmes.

Ahmad and Hossain (2015) conducted analysis of the disclosure of climate change and global warming made in the annual reports of 79 Malaysian companies. They concluded that this kind of disclosure in the annual reports of Malaysian companies is still at its introductory stage. Lipunga (2015) examined the level of CSR disclosure in the annual reports for 2012 and 2013 of Malawian quoted companies. The study indicated that the level of CSR disclosure that the companies were making in their annual reports is generally low. Particularly,

the companies were disclosing poorly on environment category. Similarly, Nurhayati, Taylor and Tower (2015) revealed that the extent of social and environmental disclosure in annual reports of Indian textile companies is low.

Vilar and Simao (2015) investigated how the banks use their web sites to disclose their social responsibility concerns and activities. The study revealed that the banks disclose on their websites on performance, environmental socioeconomic programs and other CSR information. The study also revealed that there are geographic patterns in the quantity and detail of the disclosures. The banks belong to Europe, the American continent, and Oceania, were disclosed more information. The study concluded that the disclosure of CSR by the banks is larger and more detailed according to the development level of the country where they operate.

Adopting descriptive research, Innocent et al. (2015) examined stakeholder's (investors, consumers and chartered accountants) perspective on the effectiveness of triple bottom line disclosure practices of Nigerian firms. The findings indicated that investors, consumers and chartered accountants are dissatisfied with the extent of firms TBL disclosure practice in Nigeria, and the firms` reporting was often vague and far from the expression of actual performance. Kaur (2015) explored the item wise variation among different environmental disclosure categories made by Indian companies. The study revealed insignificant differences among the environmental disclosure categories

More recently, Nurhayati et al. (2016) investigated the social and environmental reporting of Indian textile and apparel firms. The study reported a low extent of social and environmental reporting by the sample firms, with a mean disclosure of 14%, while firms reported relatively more extensive environmental information, with a mean disclosure of 18.4%. Hewaidy (2016) evaluated social and environmental disclosure practices in the annual reports of a sample of 43 companies listed in Kuwait Stock Exchange. The results revealed that the overall disclosure level for the sample companies is 21%, and the disclosure level varies by disclosure category.

In high environmentally sensitive industries, including oil and gas industries, the literature revealed also low level of quantity and quality of social and environmental disclosure. For example, Guenther et al. (2007) examined the status of environmental reporting practice of global mining, oil and gas companies. Using GRI indicators, the study analyzed 48 CSR reports for 2005. The study indicated that on average, the mining, oil and gas companies disclosed approximately 31% of the total GRI indicators (11 out of a total of 35 indicators). However, only 8% of total environmental indicators were disclosed with both high quantity and high quality. Frynas (2009) indicated that many oil companies from developing countries provide little concrete data on social and environmental issues

Ane (2012) examined the environmental disclosure quality of listed firms in heavily pollution industries (including, electricity, steel, oil chemicals, mining, etc.) in China, and indicated that the overall environmental information disclosure quality is low.

Sen et al. (2011) indicated that the voluntary environmental disclosure by oil and petrochemicals, mining and minerals, steel and cement companies in India is incomplete, more qualitative and provide inadequate disclosure for most of the environmental themes.

Oba and Fodio (2012b) investigated the extent of environmental disclosures in oil and gas and construction industries in Nigeria. The results provided evidence on the poor environmental disclosure levels in the annual reports of sampled companies. The results also indicated that the oil and gas industry provided a better disclosure level but this difference was not significant. Al-Drugi and Abdo (2012) investigated the development of environmental disclosures by oil and gas companies operating in a developing country of Libya from They revealed that although, 2002 to 2009. environmental disclosure has witnessed improvement during the period, but the level of CED is still low. Eljayash et al. (2012) examined the quantity and quality of CED in annual reports by national oil and gas companies in Middle East and North Africa (MENA), particularly Arab oil exporters. They revealed that, overall; CED in Arab oil countries is still low compared with other oil companies in developed countries.

Eltaib (2012) examined the environmental accounting disclosures of Australian oil and gas companies. Annual reports and stand-alone sustainability reports of the 10 largest Australian oil and gas companies listed in Australian Stock Exchange over the period 2005-2010 were analyzed. The results showed that environmental disclosure trend fluctuated during the study period. The results also indicated that the most of the disclosed environmental information is favourable, nonfinancial, pure narrative and general information. Summerhays and De Villiers (2012) using a sample of the largest six international oil companies examined the disclosure patterns and strategies in response to the Gulf of Mexico oil spill. The findings indicated that the overall environmental disclosures of the oil companies increased after the oil spill.

Eljayash et al. (2013) examined the differences environmental disclosure practices between national oil and gas companies and international oil and gas companies operating in Arab petroleum exporting countries. The study concluded that despite the slight increase in the environmental disclosure practices in national companies; the difference is still significant compared with international companies. Recently, Mughal (2014) examined CSR disclosure practice of petroleum companies in Pakistan. The study highlighted that petroleum companies in Pakistan are contributing positively towards CSR, more conscious towards portraying their image and they have understood the importance of disclosing environmental information other than financial information.

More Recently, Comyns and Figge (2015) explored the evolution of greenhouse gas reporting quality of 45 oil and gas companies listed on the 2011 Global Fortune 500 index. The study also investigated whether the evolution of reporting quality is linked with the type of information. This study revealed that, in total, 80 per cent of 245 reports contained quantitative and qualitative data on GHG emissions while the remaining 20 per cent

contained only qualitative data. The study also revealed that GHG reporting quality has not improved significantly between 1998 and 2010, and the type of information is important in terms of quality evolution. Eljayash (2015) investigated environmental disclosure in the oil companies in three countries of the Arab Spring (Egypt, Libya and Tunisia). The results of the study indicated low level and quality of environmental information disclosed in the annual reports before Arab spring.

Nonetheless of these results, there are some previous studies that showed high levels of environmental disclosure. For example, Yusoff, Lehman, & Nasir (2006) examined environmental disclosure and motivations among Malaysian publiclisted companies. The study revealed high levels of environmental disclosures concerning current environmental engagements and future environmental plans/strategies, and Aburaya (2012) indicated that the level of corporate environmental disclosure quality in the UK was 72.74%.

Despite the importance of disclosure quality, reporting quantity has dominated prior social and studies, environmental reporting hut overlooked the significance of what the information is being communicated. Thus, while there are a large number of studies that have addressed the social and environmental disclosure, the majority of these studies have focused on the quantity of disclosure, whereas, a scant attention has given to quality of such disclosure. This limitation of literature has noted by many previous studies. For example, Silva (2008) noted that several studies in literature overlooked reporting quality and instead confined themselves to reporting quantity. Belal and Momin (2009) also stated that most of the previous CSR (including CER) studies available from the context of developing countries are descriptive in nature and are limited to measuring the extent and volume of disclosure in annual reports.

According to Eltaib (2012) and Rupley et al. (2012), the majority of environmental disclosure research is confined to consideration of the quantity, rather than quality, of information disclosed. Aburaya (2012) stated that, research investigating issues as environmental disclosure quality is quite essential, as most previous studies concentrate on the volume of disclosure rather than its quality. Similarly, Haji (2013) noted that most previous CSR studies measured the extent of CSR disclosures and few have measured the quality of CSR disclosures. Cuesta and Valor (2013) stated that, the problem of reporting quality has been of limited interest in the literature and few studies have attempted to investigate quality of environmental, social and governance disclosure. Ahmad and Haraf (2013) also commented that the majority of prior studies related to environmental disclosure have focused on the quantity of disclosure but scant attention has given to disclosure quality. Until now, researchers still identify literature lacks regarding disclosure quality. For example, Michelon et al. (2015) stated that with the exception of a few studies those focus on disclosure quality, the CSR disclosure previous studies are not able to capture the quality of the disclosure.

However, from literature review, it is noted that, with the exception of a few studies (e.g. Aburaya, 2012; Ahmad and Haraf, 2013; Ane, 2012;

Belal, 2000; Brammer & Pavelin, 2006, 2008; Comvns and Figge, 2015; Cormier et al., 2005; Cuesta and Valor, 2013; Darus et al., 2013; Dong et al., 2015; Eakpisankit, 2012; Eljayash et al., 2012; Haji, 2013; Hassan, 2010; Harun et al., 2013; Hooks & Van Staden, 2011; Lu et al., 2015; Michelon et al., 2015; Oba and Fodio, 2012a; Rupley et al., 2012; Sulaiman et al., 2014; Wiseman, 1982), who focus on disclosure quality, the social and environmental disclosure previous studies were not able to capture the quality of the disclosure. Many authors have stressed that the quality of environmental information reported should be considered (cf. Adams et al., 1998; Clarkson et al., 2008; Hall, 2002; Silva, 2008; Sulaiman et al., 2014). This called for environmental disclosure studies dedicated to the investigation of aspects beyond the disclosure level. such as disclosure quality.

Another limitation of literature is that, many prior social and environmental disclosure studies used disclosure quantity to measure disclosure quality (Hussainey and Mouselli, 2010; Michelon et al., 2015; Ng, 1985). This may be because these studies proposed that the disclosure significance can be reflected by the disclosure quantity. However, many researchers have cautioned that much information does not mean that it has high quality, therefore, quantity or volume of information reported is not appropriate measure for reporting information quality. For example, Buzby (1975) argued that disclosure level is not the same as its sufficiency; hence, the former cannot measure the overall disclosure quality. Wiseman (1982) argued that the environmental disclosure length does not reflect its quality. Freedman and Stagliano (1992) argued that although the quantity of reporting sheds some light on the importance of information, it fails to reflect the full communicative content of the information, and as such, it is riddled with limitations in terms of a complete measurement of reporting quality. Deegan and Gordon (1996) argued that the assumption that the significance of a disclosure can be meaningfully represented by the quantity is incorrect. Similarly, KPMG (1999) suggested that disclosure quality is not synonymous with disclosure quantity.

Hussainey and Mouselli (2010) stated that disclosure quantity alone is not a satisfactory proxy to measure disclosure quality. Michelon et al. (2015) argued that the disclosure instruments used in and environmental disclosure previous social studies have been built primarily on a checklist of items that capture the amount and variety of disclosure do not sufficiently determine the quality of information. In practice, despite efforts that spent by some related organizations resulted in some standardization of corporate social environmental reporting, particularly in terms of format, but their approach to indicators is unlikely to produce high quality (Cuesta and Valor, 2013). To overcome this limitation, this study measures the quality of environmental disclosure using an environmental disclosure index and scoring scheme that able to sufficiently determine not just the quantity, but the quality of disclosure.

Moreover, most of studies related to environmental disclosure quality have concentrated on developed countries, while, there is a lack of studies addressing the quality of environmental disclosure in the developing countries. Thus, this study examines environmental disclosure quality in developing countries.

However, reviewing pertinent prior literature revealed that; the majority of previous studies have concentrated on disclosure quantity, while scant attention has given to disclosure quality. Moreover, the majority of these studies focused on a sole media of reporting (often annual report), while, a few studies have covered several reporting mediums. Most of studies related to environmental disclosure quality have concentrated on developed countries, while, there is a lack of studies addressing the quality of environmental disclosure in developing countries. And in terms of sector, there are a few studies examined environmental disclosure in oil and gas industry (cf. Alciatore and Dee, 2006; Al-Drugi and Abdo, 2012; Barr, 2007; Bose, 2006; Dibia and Onwuchekwa, 2015; Eljayash et al., 2012; Eljayash et al., 2013; Guenther et al., 2007; Heflin and Wallace, 2014; Oba and Fodio, 2012b; Patten, 1992; Summerhays and De Villiers, 2012; Sustainability Ltd. & UNEP, 1999). Moreover, with exception of Oba and Fodio (2012a) and Eljavash et al. (2012), there have been no studies done on the quality of environmental disclosure in oil and gas industry. However, these two studies are suffering from many limitations such as limiting themselves to annual reports and the samples are small. Furthermore, the samples in prior studies have tended to be small and more concentrating on developed countries. Hwoever, the prior research has also shown inconclusive results regarding the relationships between the environmental disclosure quality and some independent variables, and their relationship signs and therefore it is considered to be productive to empirically re-examine the relationships between them. Therefore, this study attempts to fill the existing gaps and overcome the limitations of the literature by the following: 1) investigating environmental disclosure quality of oil and gas companies in developing countries; 2) investigating the main reporting mediums of environmental information (namely, annual reports, stand-alone environmental reports and corporate homepages in aggregate.

3. RESEARCH METHODOLOGY

This study aims to determine the level of environmental disclosure quality of oil and gas companies in developing countries. In order to achieve the objective of this study this study adopted a quantitative research methodology and probability cluster sampling technique employed. A number of 116 oil and gas companies originated from 19 developing countries (namely, Argentina, Brazil, China, Colombia, Egypt, India, Indonesia, Kazakhstan, Kenya, Kuwait, Nigeria, Oman, Pakistan, Qatar, Republic of Korea, Saudi Arabia, Thailand, Trinidad and Tobago, and Turkey) were included in the sample. To measure quality of environmental disclosure, a 42-items disclosure was developed by adapting pertinent established indices. The study also used Wiseman's (1982) scoring scheme which was widely adopted by many pertinent studies (e.g. Cormier et al., 2004; Hughes et al., 2001; Kuo and Chen, 2013; Sulaiman et al., 2014; Zeghal & Ahmed, 1990). Using Wiseman's scoring method, the present study allocated the greatest weight (3) to quantitative disclosures environmental disclosures explained in EDI. This is followed by the next highest weight (2) that is allocated to non-quantitative but distinct information related to indicators. The lowest weight (1) is granted to general qualitative disclosures. A zero is granted to firms which did not provide information regarding a specific indicator. Thus, the total possible maximum score for the overall environmental disclosure index is 126 (i.e. 3×42= 126). The scores were converted into percentages by dividing the disclosure score of each company to the maximum possible score. The financial year ending on December 31, 2010, or June 30, 2010, or March 31, 2011, depending on the company's financial year, was chosen for the research. Annual reports, stand-alone reports and environmental related sections on homepages were downloaded from companies` websites.

4. RESULTS AND DISCUSSION

The objective of this study is to determine the level of environmental disclosure quality of oil and gas companies in developing countries. To determine the level of environmental disclosure quality, content analysis has been conducted to extract disclosure quality from annual reports, stand-alone reports, and corporate homepages of sampled companies for the year 2010. For this purpose an environmental disclosure index which adapted from various previous related studies and scoring system of Wiseman (1982) were employed. However, overall quality of environmental disclosure of the sample companies, in addition, the quality of disclosure in each categories and each reported indicators were specifically analyzed.

Annual reports, environmental stand-alone reports, environmental related sections on corporate homepages were carefully reviewed and related data extracted and coded into copies of coding sheet that has been designed for this purpose. The valid coding sheets were then entered into database of SPSS software, and then different statistical analyses were adopted. The collected data was checked for missing and outliers values. A few cases with outlier values were detected. Further checking revealed that they could not be considered unrepresentative of the population, and therefore were not excluded from the sample. In addition, goodness of data was ensured by testing data validity and reliability.

The disclosure index and scoring system were tested for reliability and validity. Cronbach's coefficient alpha is used for this purpose. The results show that Cronbach's coefficient alpha for the scale used in this study is 0.893, indicating ta high level of internal consistency for the current study's scale (Hair et al., 2007; Sekaran, 2003). This high level of internal consistency also indicates the content validity of the disclosure instrument. As it is argued that an examination of the internal consistency of the disclosure index provides some insights into the validity of the disclosure scores, as internal consistency reliability is an indirect way to test a content validity of an instrument (Sekaran 2003; Walsh, 1995). Thereafter, the data was tested using minimum, maximum, mean and standard deviation.

Table 1 displays the descriptive statistics for the environmental disclosure quality of the 116 companies in the sample. It shows the means for each of the eight index categories and overall quality of environmental disclosure. The table shows that the range of environmental disclosure quality scores varies widely, from 33 to 106. The mean score of total environmental disclosure quality per company is 68.98, which represent 54.75% out of all possible environmental disclosure scores of 126 (i.e., 42 items × maximum score of 3). This level of disclosure quality is similar to that found by Eljayash et al. (2012) who revealed that the average of quality of CED in annual reports by oil companies in the Arab oil countries was 26.66 (55.54%) in 2010. However, the level of environmental disclosure quality of the current study is relatively high compared to those found by the majority of previous studies (cf. Ahmad and Haraf, 2013; Ane, 2012; Comyns and Figge, 2015; Cuesta and Valor, 2013; Dong et al., 2015; Eakpisankit, 2012; Haji, 2013; Harun et al., 2013; Hooks & Van Staden, 2011; Michelon et al., 2015; Oba and Fodio, 2012a; Sulaiman et al., 2014).

The reason behind this relatively high quality of environmental disclosure made by oil and gas companies in developing countries is the fact that firms affiliated with more environmentally sensitive industries provided more comprehensive social and environmental disclosure than firms affiliated with less environmentally sensitive industries (Bowrin, 2013), due to the environmentally sensitive sectors receive more public scrutiny (Aburaya, 2012; Kolk and Fortanier, 2013). Another explanation is that environmental disclosures of the oil companies increased significantly in response to the spill incident which occurred from oil platforms owned by BP in the Gulf of Mexico (Eljavash et al., 2012; Summerhays and De Villiers, 2012), as the accident was an environmental crisis that not only impacted the BP image and legitimacy, but also impacted on the image and legitimacy of other oil companies (Summerhays and De Villiers, 2012). This increasing is consistent with prior research which pointed that the threats arising from specific major social and environmental incidents influence the disclosure practices of the particular corporation and other corporations within the same industry (cf. Islam and Islam, 2011; Patten, 1992, Suttipun and Stanton, 2012). Overall, given the environmentally-sensitive nature of the oil and gas industry and the increasing adverse media attention and public concern with numerous incidents associated with this industry, it is expected to observe that the quality of environmental disclosure of oil and gas industry is higher than in other industries. Also, the rationale behind this relatively high quality disclosure may be due to that the previous studies have restricted their analyzing to a single reporting medium, mostly annual report, while companies use different mediums disclosure communicate to environmental information.

Despite there are numerous companies disclose all index items, but no one of these companies

disclose full information in monetary or quantitative form. However, the maximum score obtained by the sample companies is 106 out of a possible 126, indicating significant scope for improvement even among the companies with the highest level of environmental disclosure. In addition to, the variation in the disclosure quality among the sample companies suggest that there is a need for governments and related organizations developing countries to devise more detailed guidances that specify environmental disclosure the requirements to enhance quality environmental disclosure.

Table 1 also presents the level of quality of environmental disclosure for each category. It shows that the quality of environmental disclosure of each of eight categories is different. It shows that the environmental disclosure quality of each of eight categories is different. This result is in line with some previous studies such as Hewaidy (2016) who evidenced that the disclosure level varies by disclosure category. However, analysis of the different categories showed that, the category "sustainable development" has the highest average mean of scores with 2.123, followed by "pollution abatement" (2), "health and safety" "disturbances to land and land remediation" (1.657), "environmental management" (1.629), "economic factors" (1.404), "laws and regulations" (0.97), and lastly, the category "spills & environmental incidents" has the lowest average mean (0.80).

Among all the environmental items studied, the descriptive analysis of the disclosure showed that the item "conservation of natural resources" under the category "sustainable development" represents the highest disclosure quality with mean of 2.78, followed by "air emission information" under the category "pollution abatement" and "health and safety incidents and accidents" under the category "health and safety" with means of 2.71 and 2.70 respectively. On the other hand, environmental operating costs" under the category "economic factors" represents the lowest disclosure quality with mean of $\bar{0}.28$, the second lowest item is 'costs of treatment of spills" under the category "spills environmental incidents" with mean of 0.34, and the third lowest item is "litigation" under the category "laws and regulations" with mean of 0.40.

However, from data review it was noted that for companies that scored low on the quality index did not disclose some items and/or did not disclose in monetary/quantitative terms, or did not address specific issues in their reporting. The results of this study suggest that there is a significant scope for additional and better quality environmental disclosure in various reporting mediums of oil and gas companies in developing countries. Issuing guidelines regulations and on corporate environmental disclosure is needed to motivate companies to improve their environmental disclosure.

Table 1. Descriptive Statistics of Environmental Disclosure Categories and Items

Categories	Min.	Max.	Mean	Std. Deviation
Economic Factors		I		
Past and current environmental capital expenditures	0	3	2.39	1.133
Past and current environmental operating costs	0	3	1.69	1.411
Future environmental capital expenditures	0	3	.97	1.264
Future environmental operating costs	0	3	.28	.798
Environmental liabilities and provisions	0	3	1.70	1.385
Total	0	15	7.02	4.230
Average Mean			1.404	
Laws and Regulations Litigation (present and Potential)	0	2	40	050
Fines and Penalties	0	3	.53	.959 1.008
Environmental legislations and regulations requirements	0	3	1.99	.552
Total	0	9	2.91	1.722
Average Mean	0	3	.97	1.722
Pollution abatement				
Air emission information	0	3	2.71	.528
Water discharge information	0	3	2.28	1.062
Waste disposal information	0	3	2.54	.727
Noise, odours and visual quality	0	2	.48	.839
Activities, products and services impacts on environment	1	3	2.04	.333
Installation of environmental control systems, facilities or processes described	0	3	1.92	1.040
Total	2	17	11.98	2.804
Average Mean			2	
Sustainable development				
Conservation of natural resources	0	3	2.78	.576
Recycling	0	3	1.57	1.300
Progress toward sustainability	0	3	2.52	.597
Research and development activities for sustainable development	0	3	1.63	1.026
Total	0	12	8.49	2.472
Average Mean			2.123	
Disturbances to land and land remediation		ı		•
Sites	0	3	1.54	1.106
Disturbances to land	0	3	1.71	1.111
Efforts of remediation/ Rehabilitation (present and future)	0	3	1.72	1.117
Total	0	9	4.97	2.574
Average Mean 1.657 Spills & Environmental incidents				
Number and nature of spills	0	3	1.27	1.295
Efforts to reduce and / or prevent spills	0	3	.78	1.045
Costs of treatment of spills.	0	3	.34	.835
Total	0	9	2.40	2.509
Average Mean		,	.80	2.303
Environmental management				
Environmental policies or company concern for the environment	2	3	2.06	.239
Environmental management system (EMS)	0	3	1.83	.608
Environmental auditing	0	3	1.44	1.024
Goals and targets	0	3	2.11	.872
Environmental Awards and Recognition	0	3	1.03	1.012
Department/ committee for environmental affairs pollution control	0	2	1.08	.988
Joint projects with other firms on environmental management	0	3	1.54	.888
Involvement to environmental organizations	0	3	1.31	.973
Environmental activities and programmes	0	3	2.16	.844
Environmental training and education programmes	0	3	1.72	1.139
Total	2	26	16.29	4.936
Average Mean			1.629	<u> </u>
Health and Safety				T
Employee health and safety policy	2	3	2.05	.222
Health and safety laws and regulations	0	3	1.97	.347
Health and safety management systems	0	3	1.76	.730
Health and safety at work	0	3	2.13	.880
Toxic hazard	0	3	1.11	1.070
Health and safety training	0	3	1.78	1.072
Health and safety auditing Health and safety incidents and accidents	0	3	1.39 2.70	1.011 .531
Total	6	21	14.88	3.182
Average Mean	U	41	1.86	3.102
Total	33	106	68.98	15.514
Note: N= 116			00.00	10.011

5. CONCLUSIONS

The objective of this study is to determine the level of environmental disclosure quality of oil and gas companies in the developing countries. In order to achieve this objective this study adopted a quantitative research methodology and probability cluster sampling technique was employed. A number of 116 oil and gas companies originated from 19 developing countries were included in the sample. To measure the quality of environmental disclosure, a 42-items disclosure index, scoring scheme and decision rules were developed by adapting pertinent established indices and decisions rules of prior studies. Annual reports, stand-alone reports and environmental related sections on homepages were downloaded from companies` websites.

Using the index and scoring scheme, content analysis was conducted. The data was analyzed using descriptive analysis to determine the level of disclosure quality. The results of the analysis revealed that, the total scores in annual reports, stand-alone reports and homepages range from 33 to 106 scores with an average of 68.98 scores descriptive analysis for (54.75%). The environmental disclosure quality also revealed that, the means of scores across different disclosure media are varied. Particularly, the mean of annul reports scores is 52.63, for stand-alone reports 65.64 scores and 38.53 scores for homepages. These results indicate that, there is variation in the quality of environmental disclosure among the three mediums, and suggest that stand-alone reports have the highest level of environmental discourse quality, while, the corporate homepages have the lowest

The results also revealed that the level of environmental disclosure quality differs across different categories. The category "sustainable development" has the highest level of the disclosure quality with average mean of scores 2.123, whereas, the category of "spills & environmental incidents" has the lowest level of the disclosure quality with average mean 0.80. With respect to all disclosure items, the results indicated that the item "conservation of natural resources" under the category "sustainable development" has the highest disclosure quality with mean of 2.78, while the item "future environmental operating costs" under the "economic factors" has the lowest category disclosure quality with mean of 0.28.

Prior environmental disclosure literature has not given much emphasis on disclosure quality; instead, it focused more on the quantity of disclosure. The current study seeks to fill the gap in the literature by considering the issue environmental disclosure quality (rather than quantity). Assessing the quality of the environmental reporting enables an identification of the strengths and weaknesses in current reporting practice and advances our understanding. In addition, contrary to the most available literature that only focuses on sole medium of environmental disclosure (mostly annual report), the current study contributes to the literature by covering most common vehicles of environmental disclosure, particularly, annual stand-alone reports and corporate homepages. This study also fills the void in prior disclosure literature regarding environmental whether various reporting mediums vary regarding their disclosure quality.

The present study also contributes to the environmental disclosure literature by focusing on the ED practices of specific sector (i.e. the oil and gas industry) in the DCs. It was argued that, in order to enhance our understanding on environmental disclosure behavior, it is important to focus on a specific industry (Gray et al., 1995; Ahmad and Haraf, 2013). Thus, this study contributes to

environmental disclosure literature as it provides insight into the environmental disclosure practices of oil and gas companies within developing countries, where there are limited published studies.

Understanding the ED practices of oil and gas companies enables various interested parties, such as, investors, creditors, governments, regulators and standard setter, and environmental groups to determine the quality of ED, and to assess the requirements for environmental information. It is hoped that the findings of this study serve as input towards the development of improved regulations concerning environmental reporting for the oil and gas industry, and provide guidelines to the regulators to make relevant decisions on environmental information items to be incorporated in the regulatory standards.

The findings of this study have many implications for various interested parties. The present study provides insights into environmental disclosure of a single highly environmentally sensitive industry. The study focuses on the quality of environmental disclosure in different reporting media by oil and gas companies across several countries. By assessing the quality of environmental disclosure, it enables ones to identify of the strengths and weaknesses in environmental disclosure of the sampled companies, therefore, advances our understanding of current disclosure practice by oil and gas industries in developing countries.

This study may motivate oil and gas companies in developing countries to provide environmental information in their annual reports, stand-alone reports and websites. Particularly, the findings may help the companies to focus on what should be disclosed and how to disclose. In this respect, the disclosure index serves as a guide to best practice of environmental disclosure.

In addition, by identifying the state of environmental disclosure practices, the results of this study would benefit the policy makers, regulators and reporting standards setters in proposing laws and regulations, issuing new standards, and improving environmental reporting guidelines, which in turn will lead to more transparency and better quality of environmental disclosure.

Finally, this study makes a methodological contribution to the literature by constructing an environmental disclosure quality index, which can be considered as a comprehensive enough -to some extent- and suitable for oil and gas industry, as it includes specific environmental disclosure items for this industry. Thus, the disclosure index of this study can be used as a tool for future oil and gas industry corporate environmental disclosure related research.

6. LIMITATION AND FUTURE RESEARCH

This study focused on the three main channels of environmental disclosure. As companies are using different channels and are likely using other channels to disclose environmental information, future research should investigate a wider range of those channels. Thus, besides the reporting mediums covered in this study, other common channels such as advertisements, environmental

brochure or corporate booklets, newspapers and magazines, television and radio, could be covered by future research. Moreover, future research could include reports and information that are published in other languages.

REFERENCES

- Abdalla, Y. A. and Siti-Nabiha, A. K. (2015). Pressures for Sustainability Practices in an Oil and Gas Company: Evidence from Sudan. Qualitative Research in Accounting & Management, 12) 3), 256-286. http://dx.doi.org/10.1108/QRAM-04-2014-0038
- 2. Abd Rahman, N. H., Zain, M. M. and Al-Haj, N. H. (2011). CSR Disclosures and its Determinants: Evidence from Malaysian Government Link Companies. Social Responsibility Journal, 7 (2)181-201. http://dx.doi.org/10.1108/1747111111111141486
- 3. Aburaya, R. K. (2012). The Relationship between Corporate Governance and Environmental Disclosure: UK Evidence. Unpublished PhD Thesis, Durham University, UK.
- Adams, C. A., Hill, W. Y., & Roberts, C. B. (1998) Corporate Social Reporting Practices in Western Europe: Legitimating Corporate Behaviour. The British Accounting Review, 30(1), 1-21. http://dx.doi.org/10.1006/bare.1997.0060
- 5. Ahmad, N. N. and Haraf, A. S. (2013). Environmental Disclosures of Malaysian Property Development Companies: Towards Legitimacy or Accountability? Social Responsibility Journal, 9(2), 241-258. http://dx.doi.org/10.1108/SRJ-10-2011-0090
- 6. Ahmad , N. N. and Hossain, D. M. (2015). Climate Change and Global Warming Discourses and Disclosures in the Corporate Annual Reports: A Study on the Malaysian Companies. Procedia Social and Behavioral Sciences, 172(27), 246-253. http://dx.doi.org/10.1016/j.sbspro.2015.01.361
- 7. Ahmed, N. N. and Sulaiman, M. (2004). Environmental Disclosures in Malaysian Annual Reports: A Legitimacy Theory Perspective. International Journal of Commerce and Management, 14(1), 44-58. http://dx.doi.org/10.1108/10569210480000173
- 8. Alciatore, M. L., & Dee, C. C. (2006). Environmental Disclosures in the Oil and Gas Industry. Advances in Environmental Accounting & Management, 3, 49-75. http://dx.doi.org/10.1016/S1479-3598(06)03002-0
- 9. Al-Drugi, A. and Abdo, H. (2012). Investigating the Development of Environmental Disclosures by Oil and Gas Companies Operating in Libya: A Comparative Study. International Journal of Economics and Finance Studies. 4 (2). 1-12.
- Ane, P. (2012). An Assessment of the Quality of Environmental Information Disclosure of Corporation in China. Systems Engineering Procedia, 5, 420-426. http://dx.doi.org/10.1016/j.sepro.2012.04.064
- 11. Ariweriokuma, S. (2009). The Political Economy of Oil and Gas in Africa: The Case of Nigeria. Taylor & Francis Group, London and New York.
- Barr, M. A. (2007). Corporate Social Responsibility Reporting in Emerging Economies: A Case of the Petroleum Refining Industry. Unpublished Master Thesis. Duke University.
- Belal, A. R (2000). Environmental Reporting in Developing Countries: Empirical Evidence from Bangladesh. Eco-Management and Auditing, 7(3), 114-121. http://dx.doi.org/10.1002/1099-

- 0925(200009)7:3<114::AID-EMA131>3.0.CO;2-E
- 14. Belal, A. R. (2001). A Study of Corporate Social Disclosures in Bangladesh. Managerial Auditing Journal, 16 (5), 274-289. http://dx.doi.org/10.1108/02686900110392922
- Belal, A. R. (2008). Corporate Social Responsibility (CSR) Reporting in Developing Countries: The Case of Bangladesh. Ashgate Publishers, Aldershot.
- 16. Belal, A., & Momin, M. (2009). Corporate Social Reporting (CSR) in Emerging Economies: A Review and Future Direction. Research in Accounting in Emerging Economies, 9, 119-143.
- 17. Bose, S. (2006). Environmental Accounting and Reporting in Fossil Fuel Sector: A Study on Bangladesh Oil, Gas and Mineral Corporation (Petorbangla). The Cost and Management, 34(2), 53-67.
- 18. Bowrin, A. R. (2013). Corporate Social and Environmental Reporting in the Caribbean. Social Responsibility Journal, 9 (2), 259-280. http://dx.doi.org/10.1108/SRJ-08-2011-0074
- 19. Brammer, S., & Pavelin S. (2006). Voluntary Environmental Disclosure by Large UK Companies. Journal of Business Finance & Accounting, 33(7) & (8), 1168-1188. http://dx.doi.org/10.1111/j.1468-5957.2006.00598.x
- 20. Brammer, S., & Pavelin, S. (2008). Factors Influencing the Quality of Corporate Environmental Disclosure. Business Strategy and the Environmental, 17 (2). 120-136. http://dx.doi.org/10.1002/bse.506
- Brink, P., Haines, R., Owen, S., Smith, D., & Whitaker, B. (1997). Consulting the Stakeholder: A New Approach to Environmental Reporting for IBM (UK) Ltd. In G Ledgerwood (Ed.), Greening the Boardroom: Corporate Governance and Business Sustainability (175-187). Sheffield, UK: Greenfield Publishing.
- 22. Buzby, S.L. (1975). Company Size, Listed versus Unlisted Stocks, and the Extent of Financial Disclosure. Journal of Accounting Research, 13(1), 16-37. http://dx.doi.org/10.2307/2490647
- 23. Carrots and Sticks (2013). Sustainability Reporting Policies Worldwide- Today's Best Practice, Tomorrow's Trends. 2013 ed. United Nations Environment Programme (UNEP), The Global Reporting Initiative (GRI), KPMG & The Center for Corporate Governance in Africa. Retrieved April 14, 2015, from https://www.globalreorting.org/resourcelibvrary/c
- arrots-and-sticks.pdf

 24. Chang, K. (2013). The Effects of Ownership and Capital Structure on Environmental Information Disclosure: Empirical Evidence from Chinese Listed Electric Firms. WSEAS TRANSACTIONS on SYSTEMS, 12(12), 637-649.
- 25. Chatterjee, B., & Mir, M. Z. (2008). The Current Status of Environmental Reporting by Indian Companies. Managerial Auditing Journal, 23 (6), 609-629. http://dx.doi.org/10.1108/02686900810882138
- 26. Chithambo, L. and Tauringana, V. (2014). Company Specific Determinants of Greenhouse Gases Disclosures. Journal of Applied Accounting Research, 15 (3), 323-338. http://dx.doi.org/10.1108/JAAR-11-2013-0087
- 27. Choi, J. S. (1999). An Investigation of the Initial Voluntary Environmental Disclosures Made in Korean Semi-Annual Financial Reports. Pacific Accounting Review, 11(1), 73-102.
- Clarkson, P. M., Li, Y., Richardson, G. D., & Vasvari,
 F. P. (2008). Revisiting the Relation between Environmental Performance and Environmental Disclosure: An Empirical Analysis. Accounting,

- Organizations and Society, 33(4/5), 303-327. http://dx.doi.org/10.1016/j.aos.2007.05.003
- Cormier, D., Magnan M., & Van Velthoven B. (2005). Environmental Disclosure Quality in Large German Companies: Economic Incentives, Public Pressures or Institutional Conditions?. European Accounting Review, 14(1), 3-39.
- http://dx.doi.org/10.1080/0963818042000339617 Cormier, D., Gordon, I. M., & Magnan, M. (2004). Corporate Environmental Disclosure: Contrasting Management's Perceptions with Reality. Journal of **Business** Ethics, 49(2), http://dx.doi.org/10.1023/B:BUSI.0000015844.862 06.b9
- Cuesta, M. and Valor, C. (2013). Evaluation of the Environmental, Social and Governance Information Disclosed by Spanish Listed Companies. Social Responsibility Journal, 9 (2), 220 -240. http://dx.doi.org/10.1108/SRJ-08-2011-0065
- Darus, F., Hamzah, E. A. and Yusoffa, H. (2013). CSR Web Reporting: The Influence of Ownership Structure and Mimetic Isomorphism. Procedia and Finance, http://dx.doi.org/10.1016/S2212-5671(13)00240-2
- De Villiers, C. and Van Staden, C. (2012). New Zealand Shareholder Attitudes Towards Corporate Environmental Disclosure. Pacific Accounting Review, 24 (2), 186-210. http://dx.doi.org/10.1108/01140581211258470
- Deegan, C., & Gordon, B. (1996). A Study of the Environmental Disclosure Practices of Australian Corporations. Accounting and Business Research, 26(3). 187-199 http://dx.doi.org/10.1080/00014788.1996.972951
- Deegan, C., & Rankin, M. (1996). Do Australian Companies Report Environmental Objectively? An Analysis of Environmental Disclosures by Firms Prosecuted Successfully by Environmental Protection Accounting Auditing and Accountability Journal, 9, (2), 52-69. http://dx.doi.org/10.1108/09513579610116358
- Deegan, C., Rankin, M., & Tobin, J. (2002). An Examination of the Corporate Social and Environmental Disclosures of BHP from 1983-1997: A Test of Legitimacy Theory. Accounting, Auditing & Accountability Journal, 15(3), 312-43. http://dx.doi.org/10.1108/09513570210435861
- Dibia, N. O. and Onwuchekwa, J. C. (2015). Determinants of Environmental Disclosures in Nigeria: A Case Study of Oil and Gas Companies. International Journal of Finance and Accounting, 4(3), 145-152.
- Dong, Y., Fu, R., Gao, F. and Ni, C. (2015). Determinants and Economic Consequences of Nonfinancial Disclosure Quality. Retrieved June 3, 2015. from http://papers.ssrn.1com/sol3/papers.cfm?abstrac
- t_id=2555416 M. (2015). Documentation of Eljayash, K. Environmental Disclosure Practices in the Oil Companies in the Countries of the Arab Spring -
- Some Evidences from Egypt, Libya and Tunisia. Journal of Economics, Business and Management, 3 (10), 954-960.
- http://dx.doi.org/10.7763/joebm.2015.v3.315 Eljayash, K., James, K. and Kong, E. (2012). The Quantity and Quality of Environmental Disclosure in Annual Reports of National Oil and Gas companies in Middle East and North Africa. International Journal of Economics and Finance, 4 (10), 201-217. http://dx.doi.org/10.5539/ijef.v4n10p201

- Eljayash, K., Kavanagh, M. and Kong, E. (2013). Environmental Disclosure Practices in National Oil and Gas companies and International Oil and Gas Corporations Operating in Organization of Arab Petroleum Exporting Counties. International Journal of Business, Economics and Law, 2 (1), 35-
- Eltaib, E. E. (2012). Environmental Accounting Disclosures of Australian Oil and Gas Companies. Unpublished Master Thesis. University Wollongong.
- Freedman, M., & Stagliano, A. J. (1992). European Unification, Accounting Harmonization, and Social Disclosures. The Internace Accounting, 27(2), 112-122. International Journal
- Frynas, J. G. (2009). Beyond Corporate Social Responsibility: Oil Multinationals and Social Challenges. Cambridge University Press. Retrieved from www.cambridge.org. 2009. http://dx.doi.org/10.1017/cbo9780511581540
- Gray, R., Kouchy, R., & Lavers, S. (1995). Corporate Social and Environmental Reporting: A Review of the Literature and a Longitudinal Study of U.K. Accounting, Auditing Disclosure. Accountability Journal. 8(2), 47-77. http://dx.doi.org/10.1108/09513579510146996
- Guenther, E., Hoppe, H. and Poser, C. (2007). Environmental Corporate Social Responsibility of Firms in the Mining and Oil and Gas Industries: Current Status Quo of Reporting Following GRI Guidelines. Greener Management International, 53, 7-24.
- 47. Hair, J., Money, A., Page, M., & Samouel P. (2007). Research Methods for Business. John Wiley & Sons
- Haji, A. A. (2013). Corporate Social Responsibility 48. Disclosures over Time: Evidence from Malaysia. Managerial Auditing Journal, 28 (7), 647-676. http://dx.doi.org/10.1108/MAJ-07-2012-0729
- Hall, J. A. (2002). An Exploratory Investigation into the Corporate Social Disclosure of Selected New Zealand Companies (Discussion Paper Series No. 211). Palmerston North, New Zealand: Massey University, School of Accountancy. Retrieved June 2009, from http://wwwaccountancy. massey.ac.nz/Publications/211.pdf
- Harun, N. A., Abdul Rashid, A. and Alrazi, B. (2013). Measuring the Quality of Sustainability Disclosure among the Malaysian Commercial Banks. World Applied Sciences Journal, 28, 195-201.
- Hassan, T. (2010). Corporate N. Responsibility Disclosure: An Examination of Framework of Determinants and Consequences. Unpublished PhD Thesis, Durham University, UK.
- Hasseldine, J., Salama, A. I., & Toms J. S. (2005). Quantity versus Quality: The Impact of Environmental Disclosures on the Reputations of UK Plcs. The British Accounting Review, 37, 231-248. http://dx.doi.org/10.1016/j.bar.2004.10.003
- Hewaidy, A. M. (2016). Social and Environmental Disclosure in Corporate Annual Reports: Evidence from Kuwait. International Journal of Social and Sciences Management, 3(1),http://dx.doi.org/10.3126/ijssm.v3i1.14219
- He, C. and Loftus, J. (2014). Does environmental reporting reflect environmental performance? Evidence from China. Pacific Accounting Review, 26 (1/2), 134-154. http://dx.doi.org/10.1108/PAR-07-2013-0073
- Heflin, F. and Wallace, D. (2014). The BP Oil Spill: Shareholder Wealth Effects and Environmental Disclosures. Retrieved May 17, 2015, from http://ssrn.com/abstract=1933531.

- 56. Hooks, J., & Van Staden, C.J. (2011). Evaluating Environmental Disclosures: The Relationship between Quality and Extent Measures. The British Review, 43(3), http://dx.doi.org/10.1016/j.bar.2011.06.005
- Hossain, D. M., Al Bir, A. T., Tarique, K. M. and Momen, A. (2016). Disclosure of Green Banking Issues in the Annual Reports: A Study on Bangladeshi Banks. Middle East Journal Business, 11(1), 19-30.
- Hossain, M. A., Islam, K. S., & Andrew, J. (2006). Corporate Social and Environmental Disclosure in Developing Countries: Evidence from Bangladesh. Paper presented at the Asian Pacific Conference on International Accounting Issues, Hawaii. Retrieved December 10, 2009, from http://ro.uow.edu.au/commpapers/179/.
- Hughes, S. B., Anderson, A., & Golden, S. (2001). Corporate Environmental Disclosures: Are They Determining Environmental Performance? Journal of Accounting and Public Policy, 20(3), 217-40. http://dx.doi.org/10.1016/S0278-4254(01)00031-X
- Hussainey, K., & Mouselli, S. (2010). Disclosure Quality and Stock Returns in the UK. Journal of Applied Accounting Research, 11 (2), 154 - 174. http://dx.doi.org/10.1108/09675421011069513
- IEA (2015). CO2 Emissions from Fuel Combustion (2015 edition). Retrieved November 29, 2015, from http://www.iea.org/publications
- Imam, S. (2000). Corporate Social Performance Reporting in Bangladesh. Managerial Auditing Journal, 15(3), 133-41. http://dx.doi.org/10.1108/02686900010319384
- Innocent, O. C., Gloria, O. T. and Benjamin, O. (2015). Effectiveness of Triple Bottom Line Disclosure Practice in Nigeria-Stakeholders Perspective. European Journal of Accounting Auditing and Finance Research, 3(3), 45-61.
- Islam, M. A. and Islam, M. A. (2011). Environmental Incidents in a Developing Country and Corporate Environmental Disclosures. Society and Business 6(3),http://dx.doi.org/10.1108/17465681111170984
- Islam, S., Hosen A., & Islam, M. (2005). An Corporate Examination of Environmental Disclosure by The Bangladesh Public Limited Companies. Pakistan Journal of Social Sciences, 3(9), 1095-1102.
- Joseph, C., Pilcher, R. and Taplin, R. (2014). Malaysian Local Government Internet Reporting. Sustainability Pacific Accounting Review, 26 (1/2), 75-93. http://dx.doi.org/10.1108/PAR-07-2013-0071
- Kamla, R. (2007). Critically Appreciating Social Accounting and Reporting in the Arab Middle East: Postcolonial Perspective. Advances Accounting, 20. 105-177. International http://dx.doi.org/10.1016/S0897-3660(07)20005-4
- Kamla, R. and Rammal, H. G. (2013). Social Reporting by Islamic Banks: Does Social Justice Matter? Accounting, Auditing & Accountability Journal, 26 (6), 911-945. http://dx.doi.org/10.1108/AAAJ-03-2013-1268
- Kansal, M., Joshi, M. and Batra, G. S. (2014). Determinants of corporate social responsibility disclosures: Evidence from India. Advances in incorporating Accounting, Advances International Accounting, 30 (2014) 217-229.
- Kaur, M. (2015). Checking the Item wise Variation in the Environmental Reporting Practices of Indian Companies. The Macrotheme Reviewmultidisciplinary journal of global macro trends, 4(1), 10-30.

- Khlif, H., Guidara, A. and Souissi, M. (2015). Corporate Social and Environmental Disclosure and Corporate Performance. Journal of Accounting Emerging Economies, 5(1), http://dx.doi.org/10.1108/JAEE-06-2012-0024
- Kolk, and Fortanier, F. (2013).A. Internationalization and Environmental Disclosure: The Role of Home and Host Institutions. Multinational Business Review, 21(1), 87-114.
- http://dx.doi.org/10.1108/15253831311309500 KPMG (1999). KPMG International Survey of Reporting (KPMG Environmental 1999 Environmental Consulting, DeMeern, The Netherlands).
- Kuo, L. and Chen, V. Y. (2013). Is Environmental Disclosure An Effective Strategy on Establishment of Environmental Legitimacy for Organization?. Management Decision, 51(7), 1462-1487. Decision, http://dx.doi.org/10.1108/MD-06-2012-0395
- Lipunga, A. M. (2015). Corporate Social Responsibility Reporting through the Lens of ISO 26000: A Case of Malawian Quoted Companies. International Business Research, 8(2),
- Liua, Z.G., Liu, T. T., McConkey, B.G. and Li, X. (2011). Empirical Analysis on Environmental Disclosure and Environmental Performance Level Listed Steel Companies. http://dx.doi.org/10.1016/j.egypro.2011.03.382
- Lodhia, S. K. (2006). Corporate Perceptions of Webbased Environmental Communication-Exploratory Study into Companies in the Minerals Industry. Australian Iournal of Accounting and Organizational Change, 2 (1), 74-
- http://dx.doi.org/10.1108/18325910610654135 Lu, Y. & Abeysekera, I. (2014). Stakeholders' Power, Corporate Characteristics, and Social and Environmental Disclosure: Evidence from China. Journal of Cleaner Production, 64, 426-436. http://dx.doi.org/10.1016/j.jclepro.2013.10.005
- Lu, Y., Abeysekera, I. and Cortese, C. (2015). Corporate Social Responsibility Reporting Quality, Board Characteristics and Corporate Social Reputation: Evidence from China. Pacific Accounting Review, 27(1),95-118. http://dx.doi.org/10.1108/PAR-10-2012-0053
- Matthew, M. R. (1993). Socially Responsible Accounting. Chapman & Hall. London.
- Michelon, G., Pilonato, S. and Ricceri, F. (2015). CSR Reporting Practices and the Quality of Disclosure: An Empirical Analysis. Critical Accounting, Perspectives on http://dx.doi.org/10.1016/j.cpa.2014.10.003
- Momin, M. A. and Parker, L. D. (2013). Motivations for Corporate Social Responsibility Reporting by MNC Subsidiaries in an Emerging Country: The Case of Bangladesh. The British Accounting Review, (2013),http://dx.doi.org/10.1016/j.bar.2013.06.007
- Mughal, S. L. (2014). Corporate Social Disclosure: A Case Study on Petroleum Industry of Pakistan. Journal of Economics and Sustainable Development, 5(11), 48-56.
- Neu, D., Warsame, H., & Pedwell, K. (1998). Managing Public Impressions: Environmental Disclosures in Annual Reports. Accounting, Organizations and Society, 23(3), 265-282. http://dx.doi.org/10.1016/S0361-3682(97)00008-1
- Noodezh, H. R. and Moghimi, S. (2015). Environmental Costs and Environmental Information Disclosure in the Accounting Systems. International Journal of Academic Research in Accounting, Finance and Management Sciences,

- 5(1), 13-18.
- Nurhayati, R., Taylor, G. and Tower, G. (2015). Investigating Social and Environmental Disclosure Practices by Listed Indian Textile Firms. Proceedings of the Australasian Conference on Business and Social Sciences 2015, Sydney (in partnership with The Journal of Developing Areas), p 201-212, ISBN 978-0-9925622-1-2. http://dx.doi.org/10.1353/jda.2015.0109
- Nurhayati, R., Taylor, G., Rusmin, R., Tower, G. and Chatterjee, B. (2016). Factors Determining Social and Environmental Reporting by Indian Textile and Apparel Firms: A Test of Legitimacy Theory. Social Responsibility Journal, 12(1), 167-189. http://dx.doi.org/10.1108/SRJ-06-2013-0074
- Oba, V. C. and Fodio, M. I. (2012a). Board Characteristics and the Quality of Environmental Reporting in Nigeria. Journal of Accounting and Management, 1(2), 33-48.
- Oba, V. C. and Fodio, M. I. (2012b). Comparative Analysis of Environmental Disclosures in Oil and Gas and Construction Industries in Nigeria. Journal of Sustainable Development in Africa, 14(6), 19-28.
- Patten, D. M. (1992). Intra-Industry Environmental Disclosure in Response to the Alaskan Oil Spill: A Note on Legitimacy Theory". Accounting, Organizations and Society, 17 (5), 471-475. http://dx.doi.org/10.1016/0361-3682(92)90042-Q
- Peiyuan, G. (2005). Corporate Environmental Reporting and Disclosure in China. Tsinghua University, Beijing, School of Public Policy and Management.
- Perera, M. H. B., & Matthew, M. R. (1990). The Cultural Relativity of Accounting and International Pattern of Social Accounting. Advances in International Accounting, 3, 215-251.
- Perez. A. (2015). Corporate reputation and CSR reporting to stakeholders: Gaps in the literature and future lines of research. Communications: An International Journal, 20 (1), 11-29. http://dx.doi.org/10.1108/CCIJ-01-2014-
- Rattanaphaphtham, K. and Kunsrison, R. (2011). Environmental Information Disclosure Quality, Competitive Advantage and Sustainable Growth of Thai Listed Firm: Perspective of Resource Based View. Journal of International Business and Economics, 11(3), 134-146.
- Rizk, R., Dixon R., & Woodhead A. (2008) Corporate Social and Environmental Reporting: A Survey of Disclosure Practices in Egypt. Social Responsibility 306-323. Journal, 4(3), http://dx.doi.org/10.1108/17471110810892839
- Rupley, K. H., Brown, D. and Marshall, S. (2012). Governance, Media and the Quality Environmental Disclosure. Journal of Accounting Policy, Public 31(6), 610-640. and http://dx.doi.org/10.1016/j.jaccpubpol.2012.09.00
- Said, R., Omar, N. and Abdullah, W. N. (2013). Empirical Investigations on Boards, Business Characteristics, Human Capital and Environmental Reporting. Social Responsibility Journal, 9 (4), 534-553. http://dx.doi.org/10.1108/SRJ-02-2012-0019
- Said, R., Zainuddin, Y. H. and Haron, H. (2009). The Relationship between Corporate Social Responsibility Disclosure and Corporate Governance Characteristics in Malaysian Public Listed Companies. Social Responsibility Journal, 5(2), 212 - 226. http://dx.doi.org/10.1108/17471110910964496
- Sekaran, U. (2003). Research Methods for Business: A Skill Building Approach, (4th ed.). John Wiley &

- Sons (Asia) Pte. Ltd., Singapore for manufacture and export.
- 100. Sen, M., Mukherjee, K. and Pattanayak, J. K. (2011). Corporate environmental disclosure practices in India. Journal of Applied Accounting Research, 12 (2), 139-156.http://dx.doi.org/10.1108/09675421111160709
- 101. Silva, T-A. D. (2008). Voluntary Environmental Reporting: The Why, What and How. Unpublished PhD thesis, Lincoln University.
- 102. Suchman, M. C. (1995). Managing Legitimacy: Strategic and Institutional Approaches. Academy of Management Review, 20 (3), 571-610.
- 103. Sulaiman, M., Abdullahb, N. and Fatima, A. H. (2014). Determinants of Environmental Reporting Ouality in Malaysia. International Journal of Economics, Management and Accounting, 22 (1), 63-90.
- 104. Summerhays, K. and De Villiers, C. (2012). Oil Company Annual Report Disclosure Responses to the 2010 Gulf of Mexico Oil Spill. Journal of the Asia-Pacific Centre for Environmental Accountability, 18 (2), 103-130.
- 105. Sustainability Ltd. & UNEP (1999), The Oil Sector Report: A Review of Environmental Disclosure in the Oil Industry. Retrieved August 1,2009, from http://www.sustainabilty.com/downloadspublic/i nsight/reports/oilsector.pdf
- 106. Suttipun, M. and Stanton, P. (2012). The in Corporate Environmental Differences Disclosures on Websites and in Annual Reports: A Case Study of Companies Listed in Thailand. International of Journal **Business** Management, 7 (14), 18-31. http://dx.doi.org/10.5539/ijbm.v7n14p18
- 107. United Nations Research Institute for Social Development. (2000).Promoting Responsible Business in Developing Countries: The Potential and Limits of Voluntary Initiatives.
- 108. Van Staden, C. J. and Hooks, J. (2007). A of Comparison Comprehensive Corporate Environmental Reporting and Responsiveness. British Accounting Review, 39(3), 197-210. http://dx.doi.org/10.1016/j.bar.2007.05.004
- 109. Van der Steden, W. A., Young, S. M., & Chen C. X. (2007). Doing Management Accounting Survey Research. Edited by Chapman, C. S., Hopwood, A. G. and Shields, M. D. Vol. 1, Handbook of Management Accounting Research. Amsterdam, Elsevier Ltd.
- 110. Vilar, V. H. and Simao, J. (2015). CSR Disclosure on the Web: Major Themes in the Banking Sector. International Journal of Social Economics, 42(3), 296-318. http://dx.doi.org/10.1108/IJSE-10-2013-0240
- 111. Vuorela, S. (2014). Environmental Disclosure in Finnish and Swedish Annual and sustainability Reports a Study of Forest Industry. Unpublished Bachelor Thesis in Accounting, University of Gothenburg.
- 112. Walden, W. D., & Schwartz, B. N. (1997). Environmental Disclosures and Public Policy Pressures. Journal of Accounting Public Policy, 16(2), 125-154. http://dx.doi.org/10.1016/S0278-4254(96)00015-4
- 113. Walsh, W. B. (1995). Test and Assessment, (3rd ed.). New Jersey: Prentice Hall.
- 114. Wiseman, (1982). An Evaluation I. Environmental Disclosure Made in Corporate Annual Reports. Accounting, Organizations and society. 7(1), 53-63.
 - http://dx.doi.org/10.1016/0361-3682(82)90025-3
- 115. Yin, W. H. (2012). Corporate Social Responsibility



- and Firm Performance: Evidence from China. Master Thesis, Lingnan University.
- 116. Yusoff, H. and Lehman, G. (2009). Corporate Environmental Reporting through the Lens of Semiotics. Asian Review of Accounting, 17 (3), 226-246.
- http://dx.doi.org/10.1108/13217340910991938 117. Yusoff, H and Othman, R. (2013). Environmental Reporting Practices in Malaysia and Australia. The Journal of Applied Business Research, 29(6), 1717-1726. http://dx.doi.org/10.19030/jabr.v29i6.8209
- 118. Yusoff, H., Lehman, G., & Nasir, N. M. (2006). Environmental Engagements through the Lens of Disclosure Practices: A Malaysian story. Asian Review of Accounting, 14(1/2), 122-148.

- http://dx.doi.org/10.1108/13217340610729509
- 119. Yusoff, H. and Darus, F. (2014). Mitigation of Climate Change and Prevention of Pollution Activities: Environmental Disclosure Practice in Islamic Financial Institutions. Procedia Social and Behavioral Sciences, 145, 195-203. http://dx.doi.org/10.1016/j.sbspro.2014.06.027
- 120. Zeghal, D., & Ahmed, S. A. (1990). Comparison of Social Responsibility Information Disclosure Media Used by Canadian Firms. Accounting, Auditing & Accountability Journal, 3(1), 38-53. http://dx.doi.org/10.1108/09513579010136343